

Open Research Online

The Open University's repository of research publications and other research outputs

Using Boulder Tracks as a Tool to Understand the Bearing Capacity of Permanently Shadowed Regions of the Moon

Journal Item

How to cite:

Sargeant, H.M.; Bickel, V. T.; Honniball, C. I.; Martinez, S. N.; Rogaski, A.; Bell, S. K.; Czaplinski, E. C.; Farrant, B. E.; Harrington, E. M.; Tolometti, G. D. and Kring, D. A. (2020). Using Boulder Tracks as a Tool to Understand the Bearing Capacity of Permanently Shadowed Regions of the Moon. Journal of Geophysical Research: Planets, 125(2), article no. e2019JE006157.

For guidance on citations see \underline{FAQs} .

 \odot 2020 The Authors

Version: Version of Record

Link(s) to article on publisher's website: http://dx.doi.org/doi:10.1029/2019je006157

Copyright and Moral Rights for the articles on this site are retained by the individual authors and/or other copyright owners. For more information on Open Research Online's data <u>policy</u> on reuse of materials please consult the policies page.

oro.open.ac.uk



JGR Planets

RESEARCH ARTICLE

10.1029/2019JE006157

Key Points:

- Boulder tracks were measured from orbital imagery, and the bearing capacity of regolith in permanently shadowed regions was estimated
- Regolith in permanently shadowed regions appears to be as strong as regolith found in other regions of the Moon
- Boulder track measurements could be used to remotely estimate the traversability of PSRs for future exploration rovers

Supporting Information:

Supporting Information S1

Correspondence to:

H. M. Sargeant, hannah.sargeant@open.ac.uk

Citation:

Sargeant, H. M., Bickel, V. T., Honniball, C. I., Martinez, S. N., Rogaski, A., Bell, S. K., et al. (2020). Using boulder tracks as a tool to understand the bearing capacity of permanently shadowed regions of the moon. *Journal of Geophysical Research: Planets*, *125*, e2019JE006157. https:// doi.org/10.1029/2019JE006157

Received 4 AUG 2019 Accepted 6 JAN 2020 Accepted article online 23 JAN 2020

© 2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Using Boulder Tracks as a Tool to Understand the Bearing Capacity of Permanently Shadowed Regions of the Moon

H. M. Sargeant¹, V. T. Bickel^{2,3}, C. I. Honniball⁴, S. N. Martinez⁵, A. Rogaski⁶, S. K. Bell⁷, E. C. Czaplinski⁸, B. E. Farrant⁷, E. M. Harrington⁹, G. D. Tolometti⁹, and D. A. Kring^{10,11}

¹School of Physical Sciences, The Open University, Milton Keynes, UK, ²Department Planets & Comets, Max Planck Institute for Solar System Research, Göttingen, Germany, ³Department of Earth Sciences, Swiss Federal Institute of Technology Zurich, Zurich, Switzerland, ⁴Department of Earth and Planetary Science, University of Hawai'i, Honolulu, HI, USA, ⁵Department of Earth and Environmental Sciences, Tulane University, New Orleans, LA, USA, ⁶Department of Geology and Geological Engineering, South Dakota School of Mines and Technology, Rapid City, SD, USA, ⁷School of Earth and Environmental Science, University of Manchester, Manchester, UK, ⁸Arkansas Center for Space and Planetary Sciences, University of Arkansas, USA, ⁹Department of Earth Sciences, University of Western Ontario, London, Ontario, Canada, ¹⁰Center for Lunar Science and Exploration, Lunar and Planetary Institute, USRA, Houston, TX, USA, ¹¹NASA Solar System Exploration Research Virtual Institute, Houston, TX, USA

Abstract Permanently shadowed regions (PSRs) are abundant at the lunar poles. They experience no direct sunlight and reach temperatures as low as 30 K. PSRs are of interest as evidence suggests that some may contain water ice (H_2O/OH^-), which could provide a record of the evolution of volatiles in the inner solar system. This water ice is also a critical resource for life-support systems and rocket propellant. A better understanding of mechanical properties of PSR regolith, such as its bearing capacity, will help optimize the design of future exploration rovers and landers. Thirteen boulder tracks were identified on the edge of, or inside, south polar lunar PSR enhanced imagery and used to estimate the strength of the PSR regolith at latitudes of 70° to 76° in sites with maximum annual temperatures of 65 to 210 K. PSR boulder track features are similar to those observed in highland, mare, and pyroclastic regions of the Moon, implying similar properties of the regolith. Measured features were used to estimate bearing capacity for PSR regolith at depths of ~0.28 to 4.68 m. Estimated bearing capacity values suggest that these PSRs may be somewhat stronger than highland and mare regions at depths of 0.28 to 1.00 m. Bearing capacity in these PSRs is statistically the same as those in other regions of the Moon at depths of 1.00 to 2.00 m. The results of this study can be used to infer bearing capacity as one measure for the trafficability of lower-latitude PSRs of the type measured here.

Plain Language Summary The polar regions of the Moon contain areas that never experience sunlight, known as permanently shadowed regions (PSRs). These regions are thought to contain uneven distributions of water ice deposits. To access PSRs and their possible water deposits, we must first understand the strength of the soil at these locations to safely traverse them. Thirteen boulder tracks were identified on the edge of, or within, PSRs in the lunar south polar region using images taken by the Lunar Reconnaissance Orbiter. These images were processed to enable the measurement of boulders and their associated tracks. The tracks identified within PSRs have similar appearances to those identified outside PSRs in other regions on the Moon. The strength of soil within PSRs was estimated from the measurements taken and was shown to be at least as strong as highland and mare regions of the Moon at relatively shallow depths, although the studied PSRs show no evidence for the presence of water ice. Analysis shows that PSRs of the type measured here should be able to bear rovers at depths of at least ~30 cm. In situ measurements are required to confirm and better understand the mechanical behavior of PSR regolith at shallow depths.

1. Introduction

The International Space Exploration Coordination Group (ISECG) Global Exploration Roadmap (2018) outlines the shared goals of the ISECG space agencies to expand human presence in low Earth orbit, to the Moon, and finally to Mars. A critical step in achieving these goals is continued exploration of the Moon and identifying and harvesting its available resources. Of particular interest in this endeavor are the permanently shadowed regions (PSRs) at the lunar poles. These regions are considerably colder than the rest of the

lunar surface, reaching temperatures below 30 K (Paige et al., 2010). As a result, it has been theorized that these PSRs could trap and accumulate stable water ice, as well as other volatiles, and retain them for hundreds of millions of years (Siegler et al., 2018; Watson et al., 1961). Indirect evidence and more recent direct evidence support these findings using data from the Lunar Crater Observation and Sensing Satellite (LCROSS) lunar impactor and the Moon Mineralogy Mapper, respectively (Colaprete et al., 2010; Li et al., 2018). These volatiles have a range of important applications for in situ resource utilization (ISRU), such as life-support systems and rocket propellant (e.g., Crawford, 2015). PSRs are also a valuable scientific target. The National Research Council outlined and ranked a series of scientific goals that should be considered when planning missions to the Moon (National Research Council, 2007). A future mission to a PSR would primarily address goals 4a, 4d, 7b, 7c, and 7d. These objectives pertain to the presence and behavior of volatiles on the Moon and the qualities of the regolith in which they reside. A robotic or human mission taking in situ measurements will be required to confirm volatile concentrations in the PSRs and assess the utility of any deposits for ISRU.

The success of a robotic mission is dependent on the ability of a rover to traverse the terrain. Thus, before planning a mission to a PSR, it is necessary to better understand the mechanical behavior of the regolith within these areas. Issues regarding extreme temperature lows, lack of solar energy, and static discharge (Jackson et al., 2011) in these regions are just a few of many concerns. Of particular concern for engineers is the soil behavior of the regolith in PSRs. The porosity of the regolith would influence the strength of the terrain and, thus, affect rover mobility. The porosity of PSR regolith has been inferred from the results of the LCROSS impactor where the resulting ejecta angles and flashes indicated highly porous material of ~70% to equivalent depths of 2 to 3 m (Schultz et al., 2010). Meanwhile, far-ultraviolet (FUV) reflectance properties of lunar PSRs showed FUV darkness, which could be a consequence of higher porosity, ~70%, in the surface soil compared to regions outside PSRs with porosities of ~40% (Gladstone et al., 2012). LRO Diviner indicates low thermal inertia values in the lunar high-latitude regions, which suggest that the upper few centimeters of regolith are likely highly porous (Hayne et al., 2017). Laboratory studies have also shown that low thermal cycling in PSRs could result in higher-porosity regolith (Metzger et al., 2018). As a result of these studies, one could assume that PSRs would be more difficult to traverse. If rovers and landers experience more sinkage into the regolith, becoming stuck could mean the end of a mission.

As it stands, the physical properties of the regolith in lunar polar PSRs are relatively unknown as there have been no in situ measurements taken to date. Experiments on lunar soil simulants have provided estimates of how the presence of water ice increases the soil strength (Gertsch et al., 2006; Pitcher et al., 2016). Gertsch et al. (2006) found that increasing water ice content in the simulant resulted in a stronger material, behaving like weak shale or mudstone at only 0.6% to 1.5% ice content. Similarly, the strength of icy rock has been studied and shown to be stronger than dry rock at equivalent temperatures (Atkinson et al., 2018). However, this experimental work did not provide estimates of trafficability. The following work will represent the first comprehensive study of trafficability in PSR regolith using data derived directly from lunar surface features.

2. Methodology

This study investigates the geomechanical properties of regolith within and on the edge of south polar lunar PSRs using rockfalls and their associated boulder tracks, both of which are abundant on the Moon (Bickel et al., 2019; Kumar et al., 2016). A south polar PSR map generated by McGovern et al. (2013) was used to select images, taken by the Narrow Angle Camera (NAC) on board the Lunar Reconnaissance Orbiter (LRO), of PSRs that may host a rockfall. South polar images were processed to enhance the visibility of features within the PSRs so that boulders and their tracks could be measured. Those measured values were then used to estimate bearing capacities using the methodology of Bickel et al. (2019).

2.1. Image Processing and Digital Elevation Model Processing

It is not immediately evident that boulders and boulder tracks can be discerned within shadowed regions. However, we discovered that features within PSRs can be observed using secondary light diffusely scattered from an illuminated portion of a crater containing a PSR. Adjusting the brightness and contrast of Lunar Reconnaissance Orbiter Camera (LROC) NAC images, followed by image filtering, illuminates those features sufficiently well to be measured. Image filtering is necessary because raw NAC images are affected by semisystematic vertical and horizontal stripes as well as white noise. The noise was exasperated by





Figure 1. (left) Lunar Reconnaissance Orbiter (LRO) Narrow Angle Camera (NAC) images M1117841678RE/LE contain a crater that hosts a permanently shadowed region (PSR) that is receiving a small amount of secondary sunlight being bounced from crater walls. (Middle) By altering the brightness and contrast, features within the PSR became visible. (Right) Further filtering in this work resulted in a reduction of image noise.

contrast enhancement and greatly impedes accurate measurements of boulder tracks within these areas. To suppress or remove the noise content, two separate filtering methods were implemented. We outline those methods here, while referring readers to detailed workflows and filtering information in Figure S1 and Tables S1 and S2 in the supporting information. A total of 33 LROC NAC images were downloaded and processed by either of the following methods.

The first image processing method forced an exaggeration of the dominant vertical noise in echocorrected NAC images using a sliding rectangular kernel. This technique was designed to detect and mask vertical lines with a width of a single pixel. Subsequently, a Standard Hough Transform (SHT; Hough, 1962) was applied to track the lines in the image and denote their beginning and end by using the SHT matrix. Next, these marked lines were removed from the image, and the resulting gaps in the image were interpolated using the values of the neighboring pixels. To minimize interference between adjacent lines, the filter was only run for lines that were not adjacent to another line. Once the lines were removed and interpolated, the filter was run again. This process was repeated for a specified number of times that was dependent on the noise level in the image. The resulting images were further refined with a two-pass 2-D median filter using a 3×3 pixel kernel to suppress the remaining white noise content. The filtered images were up-sampled using a bicubic interpolation in a 4×4 neighborhood to further increase image quality.

The second method was performed in ISIS3.5.1 and applied a low-pass filter kernel (1× vertical image dimension) to the echo-corrected NAC images to calculate a global column average of the images. The resulting low-frequency image was then high-pass filtered (1× horizontal image dimension), which produced a global albedo-suppressed average noise image. This image was then subtracted from the original input. Subsequently, a standard 2-D median filter within a sliding 3×3 pixel kernel was used to suppress the remaining white noise content.

Although all applied filter routines were nonaggressive, a small amount of local artifacts were observed (en échelon and cone-shaped noise patches), but they did not interfere with the performed measurements. Of these two methods, both filters were used on a case-by-case basis depending on which method produced a superior view into the PSR. Lastly, all optimized images were warped to a planetocentric polar stereographic projection and translated and exported for utilization in ArcGIS10.1. An example of a raw NAC image of a PSR and the resultant filtered image is shown in Figure 1.

Depending on local illumination conditions and spatial resolution of a NAC raw image, the quality of the filtered products was sufficient to produce digital elevation models (DEMs) within PSRs with spatial resolutions of 2 to 3 m/pixel using the Ames Stereo Pipeline (Shean et al., 2016). Details about the produced DEMs can be found in Table S3.





Figure 2. A south polar view of the Moon with permanently shadowed regions (PSRs) highlighted in red (per McGovern et al., 2013) and the location of identified PSR boulder tracks in green. The blue band area represents the region where boulder tracks were observed in PSRs, located between ~70° and 76° latitude, that is, the region where secondary illumination and image processing still allow to use NAC images without compromising spatial resolution.

2.2. Identifying Boulder Tracks

In order to see boulder tracks within a PSR, there must be some secondary illumination of the PSR when the NAC images were obtained. At latitudes ~70° to 76°, there is sufficient secondary illumination of craters so that processing of the high-resolution LRO NAC images (~0.5 m/pixel) reveals boulders and their associated tracks within the PSRs (Figure 2). Although higher-latitude PSRs have been imaged (Haruyama et al., 2008) using data from Selene's Terrain Camera and Multispectral Imager, both those instruments have spatial resolution well below that of LRO's NAC. For example, the Terrain Camera imagery used to view the floor of Shackleton crater had a resolution of 10 by 10 m, far too coarse to measure boulder shape, boulder track width, and boulder track depth. A complete survey of that geographic band was performed so that all measurable boulder tracks were identified. All boulders identified as "in shadow" in NAC images were then compared to the PSR map produced by McGovern et al. (2013), derived from ray tracing techniques applied to a topographic data set from the LRO, to confirm that the boulders lie within a PSR or on a boundary. Ultimately 11 boulder tracks were measured inside PSRs, with two boulder tracks being identified as seasonally shadowed regions, SSRs (Kloos et al., 2019). In this work, SSRs are defined as areas of shadow, which are extensions of PSR shadows, but do not fall within the PSR boundaries identified by McGovern et al. (2013). An example of a boulder track identified inside a PSR is shown in Figure 3. We recognize that secondary illumination has been mapped in PSRs closer to the pole (Cisneros et al., 2017; Koeber & Robinson, 2013; Koeber et al., 2014); however, the spatial resolution of the images has to be artificially reduced to account for the blurring introduced by the satellite's motion in combination with longer camera exposure times. Consequently, this method would not be suitable for higher-latitude PSRs with current orbital imagers.

3. Calculating Bearing Capacity

Measurements of the boulders and their associated tracks are used to estimate ultimate bearing capacity. Bickel et al. (2019) compared two methods for estimating ultimate bearing capacity based on work by





Figure 3. An example of a boulder track and boulder identified in a permanently shadowed region. The entire image is inside a permanently shadowed region that covers the crater floor and wall. The image is a detail of a filtered Narrow Angle Camera (NAC) image M1117841678R inside an unnamed crater at latitude –72.219° and longitude 154.671°. Image (a) shows the unmarked track, and image (b) shows the highlighted track.

Terzaghi (1951) and Hansen (1970). An analytical solution provided by Hansen (1970) for shallow foundations provides a better and more realistic estimate of bearing capacity, because it allows for the adaptation of the bearing capacity estimation to the local slope and boulder shape (Bickel et al., 2019). Here we apply the Hansen (1970) equation for bearing capacity, q_{f} , with the same assumptions used by Bickel et al. (2019) for other regions of the Moon:

$$q_f = cN_c s_c d_c i_c b_c g_c + q_0 N_q s_q d_q i_q b_q g_q + 0.5 \gamma B N_\gamma s_\gamma d_\gamma i_\gamma b_\gamma g_\gamma, \tag{1}$$

where *c* is the cohesion of the soil; q_0 is the vertical stress within the soil; γ is the unit weight of the soil; *B* is the width of footing; $N_{(c,q,\gamma)}$ are the bearing capacity factors, which are derived from the angle of internal friction, φ ; and $d_{(c,q,\gamma)}$ are the depth factors, $s_{(c,q,\gamma)}$ are the shape factors, $g_{(c,q,\gamma)}$ are the local slope inclination factors, and $i_{(c,q,\gamma)}$ and $b_{(c,q,\gamma)}$ are the load and foundation inclination factors, respectively. The unit weight of the soil is defined as $\rho \cdot g$, where ρ is the density of the soil and *g* is the gravity of the Moon. For a detailed explanation of the Hansen (1970) equation and its assumptions, see Bickel et al. (2019).

Density, cohesion, and internal friction angle are essential properties for understanding the behavior of the lunar regolith. These properties have been estimated through several studies; however, the values are highly varied in literature (Carrier et al., 1991; Mitchell, Houston, et al., 1972). Calculated bearing capacities in other works vary depending on the values pulled from the literature (Hovland & Mitchell, 1969; Hovland & Mitchell, 1973; Mitchell et al., 1973; Moore, 1970). Using the same conservative assumptions as Bickel et al. (2019), a cohesion value of 1 kPa (Mitchell, Houston, et al., 1972) was used for this work. As in situ measurements have yet to be taken for PSRs, highland bulk density values of 1.39 and 1.66 g/cm³ were applied for shallow (<30 cm) and deep (>30 cm) PSR soil, respectively (Mitchell, Carrier, et al., 1972), as PSRs occur





Figure 4. (a) Schematic showing specular reflection from an illuminated crater wall. (b) Schematic showing diffuse reflection from an illuminated crater wall, likely caused by large-scale roughness such as boulders and by small-scale roughness of the regolith itself. (c) Schematic of the determination of the effective incidence angle θ as a result of diffuse reflection.

in highland terranes with feldspathic composition and relatively low iron abundances (Lawrence et al., 2002; Spudis et al., 2013). The angle of internal friction (29°) was estimated from granular flow features at their static angle of repose on slopes within PSRs evaluated with high-resolution DEMs. For details on the production of those DEMs from filtered NAC images, see Table S3; for details about the determination of the static angle of repose, see Bickel et al. (2019).

4. Measurements

From the 33 NAC images that were processed, a total of 13 boulder tracks were identified in or on the edge of 5 PSRs that were found to be suitable for measurements needed to calculate the ultimate bearing capacity q_f (see Table S4). Measurements were made on the long *L* and short *B* diameters of the boulder, the width of the associated track *W*, the length of shadows cast within the track *S*, and the local slope α . All measurements were made near the boulder responsible for the track (subscript *b*) and near the beginning of the track (subscript *s*). Taking measurements at the start and end of the track allowed for the approximation of a static system, as the boulders will have little kinetic energy (Bickel et al., 2019). Thus, in general, each track produced two data points for analysis. The width and shadow length measurements were repeated three times for each of the individual tracks whenever possible in order to calculate an average and to minimize the influence of variations in track appearance, a consequence of irregular boulder shapes forming tracks of changing widths. Shadow length measurements were performed in the same area of the boulder track. Accurate measurements of the shadow length required determination of the illumination direction; the measurement was then made parallel to that direction. The measurement was most accurate when the shadow length was perpendicular to the track (Bickel et al., 2019).

Generally, the depth of a boulder track D was derived from the shadow length L and the incidence angle i of the Sun where

$$D = L \tan(i). \tag{2}$$

Determining the track depth in a PSR is challenging as the incidence angle of sunlight is not directly responsible for the length of shadows; instead, the shadow is produced from light reflected from an illuminated face of a crater. If specular reflection was wholly occurring (Figure 4a), sunlight should be reflected out of the crater. As some sunlight is clearly being reflected into the PSR, diffuse reflection is occurring (Figure 4b), likely as a result of surface roughness on these slopes.



Figure 5. Example boulder track which crosses the boundary from outside to inside a permanently shadowed region (PSR) as identified in a detail of image M1117841678LE/RE. Image processing allowed the identification of the track within the shadowed region. The PSR boundary is defined by McGovern et al. (2013). No obvious changes to track morphology were identified as the track enters the PSR.

Assuming that diffuse reflection results in a broad beam of light from the illuminated crater wall into the PSR, the effective incidence angle θ can be derived (Figure 4c). The illumination source was taken as the midpoint of the illuminated section of the crater wall H/2, which was derived from a 100-m resolution LRO Wide Angle Camera (WAC) stereo-derived DEM. Using the elevation difference between this reflecting surface and the location of the boulder h, as well as the horizontal distance between these points x, the effective incidence angle can be calculated:

$$\theta = \tan\left(\frac{x}{\left(\frac{H}{2} - h\right)}\right). \tag{3}$$

The boulder track depth D in PSRs can therefore be calculated as follows:

$$D = L\tan(\theta). \tag{4}$$

5. Results

We began with a qualitative study of PSR boulder tracks in NAC images. Boulder tracks that crossed the boundary from sunlit regions into PSRs (e.g., M1117841678LE/RE) show no significant difference in track morphology, suggesting similar geomechanical properties (Figure 5). Additionally, when compared to the tracks identified by Bickel et al. (2019) in highland, mare, and pyroclastic regions (Figure 6), the tracks do not show any significant difference in morphology. Boulders rolling into PSRs penetrate sufficiently deep to leave measurable tracks but do not completely sink into the regolith as might be expected if the regolith was unusually porous and weak. If there is a large density contrast between sunlit and PSRs, then the track size would be significantly different. Yet no difference is observed.





Figure 6. Example boulder tracks as seen in (a) highland regions, (b) mare regions, (c) pyroclastic regions, and (d) permanently shadowed regions. The highland, mare, and pyroclastic images were taken from Bickel et al. (2019). A qualitative analysis of these tracks shows that there is no significant difference in mechanical properties of permanently shadowed region regolith compared to the other regions considered; that is, the boulders do not sink particularly deeply into the regolith.



Figure 7. Estimated bearing capacity values from permanently shadowed regions are shown here as a function of slope angle. The data are separated by the track depth at each measurement (≤ 1 m, 1–2 m, and 2–6 m) as in Bickel et al. (2019). Slope traverse values are taken from the final report of the Lunar Exploration Science Working Group (LExSWG, 1995).

Once all measurements were obtained, Hansen's bearing capacity formula (equation (1)) was applied. Estimated bearing capacity values for each terrane (PSRs, highland, mare, and pyroclastics) were plotted with respect to the slope angle where the measurements were taken. The data are presented as distinct ranges for depth of track (≤ 1 m, 1 to 2 m, and 2 to 6 m) comparable to those of Bickel et al. (2019). A least squares fit was applied with a shaded error bar derived from an estimate of the standard deviation of the error (Figure 7). Potential slope limits are also shown. Bearing capacity tends to decrease as the slope angle increases, a result of the reduced soil volume bearing the boulder (Castelli & Lentini, 2012; Meyerhof, 1957) and potentially a reduction of physical soil strength due to slow downslope movement (Bickel et al., 2019; Carrier et al., 1991). In addition, the bearing capacity is shown to be higher in deeper material, which would be expected as the vertical stress and bearing soil volume increases over depth. Generally all the regions plotted follow a similar trend.

A plot of bearing capacity with depth is shown in Figure 8 for PSRs compared to highland, mare, and pyroclastic data from Bickel et al. (2019). As in Figure 7, Figure 8 also shows that the bearing capacity of the regolith appears to increase with depth, which is consistent with Apollo data (Mitchell et al., 1973). Calculated bearing capacities increase with depth





Figure 8. Estimated bearing capacity values for each terrane are presented as a function of track depth. Linear fits are only indicative (for comparison between the terranes); they do not represent the actual bearing capacity values at shallow depth, as they do not consider slope angle.

for all terranes, with PSR and pyroclastic regolith being generally somewhat stronger than highland and mare regolith at equivalent depths. However, there also appears to be a lot of scatter in the bearing capacities and overlap of values among terranes, which might be connected to varying slope angles at the respective measurement locations.

The trends produced in Figure 7 were used to estimate bearing capacity for each associated depth range as a function of slope (Table 1). The uppermost regolith that has been measured with this technique in PSRs (0.28 to 1.00 m) appears to have higher average bearing capacities than highland and mare material at ≤ 1 m but ranges overlap. The greatest variability in bearing capacity at shallow depths lies with PSRs, while at greater depths the greatest variability lies with pyroclastics. The data obtained from the boulder tracks in SSRs fit the same trend as those obtained from PSRs, with no noticeable anomalies, indicating that the soil strength properties in the SSRs and PSRs measured at these 70–76° latitude locations are similar.

A statistical analysis of the spread of calculated bearing capacities over depth was performed to determine if there are significant differences between terranes (see Table S5). A two-sample *T* test assuming unequal variances (two-tail) was applied to the calculated PSR values from this work and each data set from highland, mare, and pyroclastic calculations from Bickel et al. (2019) for the \leq 1- and 1 to 2-m depth range providing *p*

values based on a 95% confidence interval. The 2- to 6-m depth range could not be considered as there were too few data points for PSRs at this depth. Two analyses were performed, the first considered all bearing capacity values in the depth range of ≤ 1 and 1 to 2 m across all slope angles. However, as there are no data for relatively shallow slope angles in the ≤ 1 -m depth range in the highland and mare regions, a second *T* test was performed for bearing capacity data that are obtained at slope angles of $>15^{\circ}$ only for the ≤ 1 -m depth range. This allowed for a comparison of shallow data without a bias associated with slope angle dependence.

Table 1

Estimated Bearing Capacity Values and Associated Uncertainty (as Derived From the Distribution of the Data) for Each Location Type at the Specified Depth Ranges of $\leq 1 m$, 1-2 m, and 2-6 m, and for Slope Values of 0, 10, 20, and 30

	Denth of	Bearing capacity using Hansen's method at a range of slope angles (kN/m^2)			
Location	track (m)	0 °	10 °	20 °	30 °
PSRs	$\leq 1 (n = 8)$	127 ± 29	84 ± 22	41 ± 24	_
	1-2 (n = 13)	194 ± 29	134 <u>+</u> 26	75 ± 26	16 <u>+</u> 29
	2-6 (n = 5)	271 ± 63	192 ± 53	112 ± 55	32 ± 68
Highlands	$\leq 1 (n = 12)$	52 ± 14	39 ± 10	25 ± 7	11 ± 7
	1-2 (n = 17)	166 ± 24	119 ± 21	72 ± 20	24 ± 22
	2-6 (n = 51)	314 ± 63	228 ± 58	143 ± 57	57 ± 59
Mare	$\leq 1 (n = 15)$	85 ± 12	57 <u>+</u> 9	29 ± 7	1 ± 9
	1-2(n = 30)	167 <u>+</u> 19	108 ± 18	50 ± 18	_
	2-6 (n = 35)	276 ± 50	201 ± 47	126 ± 47	51 ± 50
Pyroclastics	$\leq 1 (n = 22)$	131 ± 21	86 <u>±</u> 20	41 ± 20	—
	1-2 (n = 28)	200 ± 30	144 ± 28	89 <u>+</u> 28	33 ± 30
	2-6 (n = 32)	336 ± 79	261 ± 74	18673	111 ± 78

Note. The highland, mare, and pyroclastic data were obtained from Bickel et al. (2019). Data were omitted where the estimated bearing capacity values fall below 0 kN/m² or where a trend line with uncertainty could not be obtained (\leq 3 data points). Due to Hansen's (1970) assumptions, estimated bearing capacity approaches zero when local slope angles approach the angle of internal friction; thus, data are sparse in the 30° bin.

The analysis showed that for all data in the depth range of ≤ 1 m, the estimated bearing capacity in PSRs, as determined by the distribution of data, is statistically similar to lunar pyroclastic deposits, and statistically stronger than both highland and mare regions. However, the analysis performed in the >15° slope angle range indicated that PSRs are statistically stronger than pyroclastic deposits, highland, and mare regolith. Meanwhile, at the depth range of 1 to 2 m, PSRs have statistically similar bearing capacities to pyroclastic deposits, highland, and mare regions when considering all slope angles and slope angles of $>15^{\circ}$ only. In contrast, there is some indication that the bearing capacity is weaker at greater depths. For example, an estimated value of 271 $\mathrm{kN/m^2}$ at 2 to 6 m in a PSR is lower than 314 kN/m^2 in highlands and 336 kN/m^2 in pyroclastic material at that same depth. We note, however, that there are not sufficient measurements from those depths to be statistically significant. It is important to note that there is considerable variation in any single terrane at all depths.

6. Discussion

6.1. Likelihood of Ice in Lower-Latitude PSRs

Models indicate that water ice will be stable and form on the surface of PSRs at temperatures below ~100 to 112 K (Hayne et al., 2015; Paige et al., 1992; Siegler et al., 2015; Zhang & Paige, 2009, 2010). To understand whether water ice could be present in the PSRs studied here, a maximum (~240 m/pixel) temperature map derived from the Diviner Radiometer

data (Williams et al., 2017) was evaluated. The PSRs in this study have maximum surface temperatures of ~65 to 210 K (see Table S6). Four of the boulders studied in this work, distributed between two PSRs, are found in locations that have temperatures that would be suitable for surficial surface ice to be stable, although it is not observed. The presence of CO_2 ice is unlikely in the PSRs studied here as they require temperatures of <55 K to be stable (Hayne et al., 2019; Zhang & Paige, 2009) and, therefore, these "warmer" PSRs may be more desirable targets for those wishing to obtain water ice without CO_2 impurities. A lunar polar ice map derived from Moon Mineralogy Mapper reflectance spectra (Li et al., 2018) suggests that surficial water ice (within a few millimeters of the surface) is not present at any of the studied boulder locations, suggesting that if ice has formed in the past, it would likely have redistributed deeper as a result of gardening of the upper regolith (Hurley et al., 2012). It should be noted that the PSRs studied here do not represent the PSRs found at higher latitudes that do have evidence to suggest surficial water ice.

The absence of surficial ice does not preclude the presence of subsurface ice. Water can thermally migrate through the regolith where it may remain trapped (Schorghofer & Taylor, 2007). The Oxford 3D Thermophysical Model (Warren et al., 2019) was applied to the PSRs studied in this work to estimate the temperature of the regolith with depth. The model analyzes a grid at each location with 8 ppd in the longitude direction and 16 ppd in the latitude direction. If the maximum temperature below the surface remains below 112 K, it is assumed that stable water ice could be found. The results show that water ice can be found at depths below 0.05 m in one PSR, and below 0.10 m in three further PSRs in this study (see Figure 5 and Table S6). The remaining PSR in this study was too small for the Oxford 3D Thermophysical model to be applied without the illuminated regions being included in the analysis. The estimated ice stability depths combined with the bearing capacity data suggest that low-latitude PSRs that have the potential to contain buried water are likely traversable. Further work will be needed to assess bearing capacities and, thus, trafficability in the higher latitude, colder PSRs. In situ analyses are also needed to confirm whether water ice is present at such locations and how it is distributed.

6.2. Technique Limitations

It is important to understand the limitations of the measurements and the application of the Hansen (1970) bearing capacity equation, which were discussed in detail in Bickel et al. (2019). A number of assumptions were made when determining the effective incidence angle, θ , of secondary illumination of PSRs, which affects estimated track depths and bearing capacity (see Figure S3). The uncertainty in θ is not well defined, but to illustrate the consequences of that uncertainty, an error analysis was performed assuming 5° and 10° errors in θ . For depths <6 m, a 5° and 10° uncertainty in θ results in a maximum uncertainty in our calculated bearing capacity values of q_{f-95}^{+107} kN/m² and q_{f-51}^{+59} kN/m², respectively. The largest impact of an uncertainty in θ occurs when measurements were taken at low slope angles, where the resulting θ is small. This is a consequence of the trigonometric nature of the θ estimations (see Figure 4c).

Two additional sources of potential uncertainty lie with the track depth estimations. (1) When estimating track depth, a first-order approximation was made, which assumes a flat-bottomed track. However, boulder tracks are likely to be irregular in shape, particularly as lunar boulders are particularly angular (Bickel et al., 2019; Kumar et al., 2016). As a consequence, the track depth estimations may be shallower than the actual tracks. (2) At very high incidence angles, the rim of boulder tracks might cast a very long shadow that covers a majority of the track. These long shadows do not allow for an estimate of depth at the center of the track as desired, but toward the opposite rim, ultimately resulting in inaccurate track depth estimates. The consequence is a bearing capacity that is associated with a track on average 23% deeper than originally thought.

6.3. Implications for PSR Regolith Processes

Boulder track depths measured in this work range from 0.28 to 4.68 m, and, therefore, the estimated bearing capacity results are only directly applicable in that range. When comparing the measurements reported here in PSRs with those in mare and highland regions (Bickel et al., 2019), the bearing capacities are statistically the same in all regions in the 1- to 2-m depth, which indicates that regolith processes at this depth are unaffected by temperature differences at the surface. The bearing capacities are statistically higher for PSRs at the shallowest depths (≤ 1 m) as compared to highland and mare regions. Interestingly, the bearing capacities in the PSRs at ≤ 1 m are similar to those in pyroclastic deposits at equivalent

depths. One might posit that extra strength is due to surficial or cohesive intergrain ice present in PSRs. Although remote sensing evidence suggests that there is no surficial water ice, there could be ice at depths of >10 cm as suggested by the Oxford 3D Thermophysical model. Water could have migrated beneath the surface as a consequence of impact bombardment and regolith overturn (Hurley et al., 2012). Providing water has migrated deeply enough into the regolith, perhaps the presence of water ice connects the grains forming a rigid matrix structure. If the PSRs studied in this work do not contain water ice, then an explanation is required for the statistically higher bearing capacity of PSR regolith in the upper meter. The specific location of measurements may be a factor in the increased PSR regolith strength calculated using this method. It is presumed that PSR regolith is porous; however, this study considers boulders and tracks, which are generally located on a slope. Slow downslope movement of regolith over time (see Bickel et al., 2019; Carrier et al., 1991) might have resulted in a modified porosity and density of the regolith on the slopes. These modified properties might vary from those at or closer to the center of PSRs. In situ measurements would be required to understand how representative the bearing capacity results are as compared to other locations within and around PSRs at different latitudes.

If subsurface ice existed at relatively shallow depths when the tracks were produced, the ice may have ablated from the walls of the tracks. Based on studies of boulder tracks elsewhere (Hurwitz & Kring, 2016; Kumar et al., 2016), tracks are obliterated by regolith processes on time scales of 20 to 35 Ma, although icy regolith may result in different track aging rates. Experimental data suggest that ice sublimates from hydrated lunar regolith simulants at a rate of $\leq 2\%$ in 5 days at temperatures of ~100 K (Piquette et al., 2017). Therefore, any ice exposed by the rolling boulders may have ablated before the tracks were observed by LROC.

The dimensions of the boulders measured in this work range from 4.2 to 16.3 m. This is a significantly smaller range than that found in highland, mare, and pyroclastic regions (Bickel et al., 2019), which contain boulders of 1 to 31 m in diameter (see Figure S2). The lack of small boulders in PSRs may be an artifact of image noise, which makes it difficult to identify small features. The reason for a smaller proportion of larger boulders, relative to that seen in mare and highland regions, is less clear.

6.4. Implications for PSR Traverses

The observations indicate that rovers with wheel radii of $\gtrsim ~28$ cm should be able to traverse PSRs of the type studied here. It remains possible that the upper few centimeters of regolith are physically different than deeper regolith analyzed in this study. For example, the upper few centimeters of lunar regolith are generally very porous at ~52% (Carrier et al., 1991); however, if water ice is present, then the porosity could be decreased if the ice fills the pore spaces. Additional work may, therefore, be needed to determine if vehicles with wheel diameters \$28 cm can traverse these regions. The comparative assessment of average PSR bearing capacities with those of other terranes used highland and mare measurements obtained mostly at high slope angles $(>15^{\circ})$. Because slope affects bearing capacity, that may have produced a bias toward lower bearing capacity values in highland and mare terranes. That prompts a second comparison of PSR properties with those of highland and mare using only those observations made on slopes $>15^{\circ}$ (see Table S5). On those surfaces, PSR regolith appears to be statistically stronger in the \leq 1-m range, with the caveat that the result is based on only two data points. Both comparative approaches have weaknesses but point to similar conclusions: that PSRs appear to be stronger than highland and mare regions at equivalent depths and slope angles. When comparing the results of PSRs with pyroclastic deposits, pyroclastic deposits appear to be statistically similar in strength to PSRs when all slope angles are considered, but statistically weaker when considering higher slope angles only. As the pyroclastic data have a less variable distribution across all slope angles, it is more appropriate to consider the statistical analysis for all slope angles, which indicates that pyroclastic deposits are similar in strength to PSRs at ≤ 1 m. Despite those trends, it is also true that there is a range of bearing capacities in any terrane.

6.5. Implications of Regolith Relative Density on Bearing Capacity

The value for bulk density of PSR regolith used in this work (1.66 g/cm³ for regolith >30 cm deep) was selected because the lunar polar regolith is thought to be mostly feldspathic "highland-like" material. Therefore, the bulk density value is taken from Mitchell, Carrier, et al. (1972), which analyzed highland material. However, it has also been theorized that PSRs are highly porous (Gladstone et al., 2012), which



would equate to a lower relative and bulk density. Bulk density, ρ (g/cm³), and porosity, *n*, are related as follows (Carrier et al., 1991):

$$\rho = G\rho_w(1-n),\tag{5}$$

where G is the specific gravity of regolith taken to be 3.1 and ρ_w is the density of water (1 g/cm³). When using a bulk density of 1.66 g/cm^3 , the estimated porosity would therefore be ~46%. As evidence suggests that PSRs could have porosities as high as 70% (Gladstone et al., 2012), the equivalent bulk density is calculated to be 0.93 g/cm³. As illustrated by Carrier et al. (1991) using a basaltic regolith simulant, an increase in porosity and a decrease in relative and bulk density will result in reduced cohesion and friction angle values of the respective soil. However, as the input parameters for cohesion and friction angle adopted from Bickel et al. (2019) are already conservative assumptions, particularly the angle of internal friction (29°), a further reduction would likely result in unrealistic bearing capacity estimates. A comparison of the bearing capacities as calculated using this lower-limit value for bulk density is shown in Figure S4 and Tables S7 and S8. The results show that the bearing capacity is reduced at higher porosities (e.g., from 127 ± 29 to 73 ± 14 kN/m² in PSRs at depths of ≤ 1 m and on a 0° slope for densities of 1.66 and 0.93 g/cm³, respectively); however, the upper meter of regolith in PSRs is still estimated to be as strong as highland regolith and equally as strong as mare and pyroclastic regolith. At depths >1 m, the bearing capacity of regolith in PSRs is calculated to be significantly lower (~34-45% lower) than highland, mare, and pyroclastic regolith for the lowest bulk density scenario; for example, PSRs have a bearing capacity of 110 ± 16 kN/m² when applying a density of 0.93 g/cm³ as compared to bearing capacity values calculated for highland, mare, and pyroclastic regions of $166 \pm 24, 167 \pm 19$, and $200 \pm 30 \text{ kN/m}^2$, respectively, at depths of $\leq 1 \text{ m}$ and on a 0° slope. As a consequence, the PSR regolith may be somewhat weaker at depths of >1 m, as compared to other regions of the Moon.

6.6. Requirements for Increasing the PSR Boulder Data Set

Although the data support a self-consistent set of conclusions, we recognize that it needs to be confirmed. Now that the methodology has been demonstrated, the data set could be expanded. In this work, 13 boulders across five locations were analyzed. It seems reasonable that more boulders and boulder tracks can be located and measured within the north polar region in a similar latitude range of ~70–75°. That will allow an assessment of heterogeneities that may exist in PSRs of different ages, composed of different lithologies. Some progress should be possible with the existing set of LROC NAC images, but we envision that a number of boulder tracks will be captured in some of the darkest and coldest PSRs once ShadowCam is deployed (Robinson & ShadowCam Team, 2018). However, as the spatial resolution of ShadowCam will be lower than the LROC imager, it will likely only be able to resolve large boulder tracks, which cannot be used to estimate bearing capacity at shallow depths.

7. Conclusions

A method was developed to study boulder tracks in NAC images of lunar south polar PSRs from which the bearing capacities of PSR regolith could be assessed qualitatively and quantitatively. Boulders that traveled from sunlit regions into PSRs are seemingly unaffected. They do not disappear into an overly porous or weak regolith. Rather, the dimensions of the boulder tracks remain unchanged suggesting that there are no large bearing capacity contrasts between the regolith in sunlit and permanently shaded regions. Those track dimensions can also be used to estimate bearing capacities. When compared with bearing capacities determined in other terranes (Bickel et al., 2019), regolith in PSRs appear to be as strong as the regolith in highland and mare regions, and similar to that in pyroclastic deposits within the upper ~0.28 to 1 m of regolith across all slope angles. Those qualitative and quantitative results seem to contradict reports of very high porosity in PSRs (e.g., Gladstone et al., 2012; Hayne et al., 2017; Metzger et al., 2018; Schultz et al., 2010) or require those proposed high-porosity conditions be limited to the uppermost 28 cm or the presence of water ice provides additional strength in a porous regolith. In either case, trafficability of PSRs may be possible with rovers of wheel radii \gtrsim 28 cm with sufficient traction. In situ analyses are still required to verify these results and to establish the distribution of regolith strength within the uppermost centimeters and meters, and the lateral heterogeneity of the regolith, of PSRs.



Acknowledgments

The authors would like to acknowledge the support of the Lunar and Planetary Institute (LPI) in Houston, TX, and the Universities Space Research Association. Funding was provided by the NASA Solar System Exploration Research Virtual Institute (NNXA14AB07A, PI David A, Kring). The research was performed as part of the Exploration Science Summer Intern Program hosted by the LPI and the NASA Johnson Space Center in Houston, TX. The authors would like to thank Julie D. Stopar of the LPI for her willingness to provide insights about the LROC system. The authors would also like to thank Tristram Warren of Oxford University for his work with the Oxford 3D Thermophysical model and its application to the PSRs studied in this work. H. S. would like to acknowledge the support of the Science and Technology Facilities Council studentship grant (Grant ST/N50421X/ 1). V. T. B. gratefully acknowledges the financial support by the International Max Planck Research School at the Max Planck Institute for Solar System Research at the University of Goettingen (IMPRS) and the Engineering Geology group, Department of Earth Sciences, Swiss Federal Institute of Technology Zurich (ETHZ). This research has made use of the USGS Integrated Software for Imagers and Spectrometers (ISIS) and the Ames Stereo Pipeline (ASP). The described PSR image filtering routines of method one, that is, the line exaggeration, the SHT, and the "fillmissing" algorithm, have been written and/or implemented in Matlab R2018a. Used data can be retrieved from Sargeant et al. (2019), the cited references, or from the LROC image archive (http://wms.lroc.asu.edu/lroc/ search). All used NAC image IDs can be found in Table S4. This is LPI Contribution No. 2247. LPI is operated by USRA under a cooperative agreement with the Science Mission Directorate of the National Aeronautics and Space Administration. The study was conceived by D. A. K and implemented by H. M. S., V. T. B., C. I. H., S. N. M., and A. R. with D. A. K. ISRU issues were addressed by C. I. H., A. R., and H. M. S., while V. T. B., C. I. H., S. N. M., A. R., and H. M. S. selected sites of interest. V. T. B. processed imagery and DEMs; V. T. B. and H. M. S. conceived the implementation of bearing capacity equations; S. N. M. and A. R. built the GIS and performed measurements. H. M. S., V. T. B., C. I. H., S. N. M., and A. R. analyzed the results and wrote the paper. All authors contributed to a discussion of findings, their implications for future missions, and editing the paper.

References

- Atkinson, J., Durham, W. B., & Seager, S. (2018). The strength of ice-saturated extraterrestrial rock analogs. *Icarus*, 315, 61–68. https://doi. org/10.1016/j.icarus.2018.06.016
- Bickel, V. T., Honniball, C. I., Martinez, S. N., Rogaski, A., Sargeant, H. M., Bell, S. K., et al. (2019). Analysis of Lunar Boulder Tracks: Implications for trafficability of pyroclastic deposits. *Journal of Geophysical Research, Planets*, 124, 1296–1314. https://doi.org/10.1029/ 2018JE005876
- Carrier, W. D., Olhoeft, G. R., & Mendell, W. (1991). Physical properties of the lunar surface. In G. H. Heiken, D. T. Vaniman, & B. M. French (Eds.), *Lunar Sourcebook*, (pp. 475–594). Cambridge, UK: Cambridge University Press.
- Castelli, F., & Lentini, V. (2012). Evaluation of the bearing capacity of footings on slopes. International Journal of Physical Modelling in Geotechnics, 12(3), 112–118.
- Cisneros, E., Awumah, A., Brown, H., Martin, A., Paris, K., Povilaitis, R., et al. the LROC Team (2017). Lunar Reconnaissance Orbiter Camera Permanently Shadowed Region Imaging--Atlas and Controlled Mosaics. Paper presented at the Lunar and Planetary Science Conference.
- Colaprete, A., Schultz, P., Heldmann, J., Wooden, D., Shirley, M., Ennico, K., et al. (2010). Detection of water in the LCROSS ejecta plume. *Science*, 330(6003), 463–468. https://doi.org/10.1126/science.1186986
- Crawford, I. A. (2015). Lunar resources: A review. Progress in Physical Geography, 39, 137–167. https://doi.org/10.1177/0309133314567585

Gertsch, L., Gustafson, R., & Gertsch, R. (2006). Effect of water ice content on excavatability of lunar regolith. Paper presented at AIP Conference Proceedings, International Forum on Space Technology and Applications: Embarking on the New Age of Exploration Together, Albuquerque, NM. https://doi.org/10.1063/1.2169290

- Gladstone, G. R., Retherford, K. D., Egan, A. F., Kaufmann, D. E., Miles, P. F., Parker, J. W., et al. (2012). Far-ultraviolet reflectance properties of the Moon's permanently shadowed regions. *Journal of Geophysical Research*, 117, E00H04. https://doi.org/10.1029/ 2011JE003913
- Hansen, J. B. (1970). A revised and extended formula for bearing capacity. Copenhagen: Geoteknisk Institut.
- Haruyama, J., Ohtake, M., Matsunaga, T., Morota, T., Honda, C., Yokota, Y., et al. (2008). Lack of exposed ice inside lunar south pole Shackleton crater. *Science*, 322(5903), 938–939. https://doi.org/10.1126/science.1164020
- Hayne, P. O., Hendrix, A., Sefton-Nash, E., Siegler, M. A., Lucey, P. G., Retherford, K. D., et al. (2015). Evidence for exposed water ice in the Moon's south polar regions from Lunar Reconnaissance Orbiter ultraviolet albedo and temperature measurements. *Icarus*, 255, 58-69. https://doi.org/10.1016/j.icarus.2015.03.032
- Hayne, P., Siegler, M., Paige, D., Lucey, P., & Fisher, E. (2019). Carbon dioxide frost at the poles of the Moon: Thermal stability and observational evidence from the Lunar Reconnaissance Orbiter. Paper presented at the Lunar and Planetary Science Conference, Houston, Tx.
- Hayne, P. O., Bandfield, J. L., Siegler, M. A., Vasavada, A. R., Ghent, R. R., Williams, J. P., et al. (2017). Global regolith thermophysical properties of the Moon from the Diviner Lunar Radiometer Experiment. *Journal of Geophysical Research, Planets*, 122, 2371–2400. https://doi.org/10.1002/2017JE005387
- Hough, P. V. (1962). America Patent No. 3069654. Retrieved from https://patents.google.com/patent/US3069654A/en
- Hovland, H. J., & Mitchell, J. K. (1969). Friction angle of lunar surface soils estimated from boulder tracks: Space Sciences Laboratory. Lunar Surface Engineering Properties Experiment Definition, Second Quarterly Report, Univ. of California.
- Hovland, H. J., & Mitchell, J. K. (1973). Boulder tracks and nature of lunar soil. *The Moon*, 6(1-2), 164–175. https://doi.org/10.1007/ BF02630660
- Hurley, D. M., Lawrence, D. J., Bussey, D. B. J., Vondrak, R. R., Elphic, R. C., & Gladstone, G. R. (2012). Two-dimensional distribution of volatiles in the lunar regolith from space weathering simulations. *Geophysical Research Letters*, 39, L09203. https://doi.org/10.1029/ 2012GL051105
- Hurwitz, D., & Kring, D. A. (2016). Identifying the geologic context of Apollo 17 impact melt breccias. *Earth and Planetary Science Letters*, 436, 64–70. https://doi.org/10.1016/j.epsl.2015.12.032
- Jackson, T. L., Farrell, W. M., Killen, R. M., Delory, G. T., Halekas, J. S., & Stubbs, T. J. (2011). Discharging of roving objects in the lunar polar regions. Journal of Spacecraft and Rockets, 48(4), 700–704. https://doi.org/10.2514/1.51897
- Kloos, J. L., Moores, J. E., Sangha, J., Nguyen, T. G., & Schorghofer, N. (2019). The temporal and geographic extent of seasonal cold trapping on the Moon. Journal of Geophysical Research, Planets, 124, 1935–1944. https://doi.org/10.1029/2019JE006003
- Koeber, S., & Robinson, M. (2013). LROC observations of permanently shadowed regions. Paper presented at the Lunar and Planetary Science Conference, Houston, Tx.
- Koeber, S., Robinson, M., & Speyerer, E. (2014). LROC observations of permanently shadowed regions on the Moon. Paper presented at the Lunar and Planetary Science Conference, Houston, Tx.
- Kumar, P. S., Sruthi, U., Krishna, N., Lakshmi, K., Menon, R., Krishna, B. G., et al. (2016). Recent shallow moonquake and impact-triggered boulder falls on the Moon: New insights from the Schrödinger basin. *Journal of Geophysical Research, Planets*, 121, 147–179. https://doi. org/10.1002/2015JE004850
- Lawrence, D. J., Feldman, W. C., Elphic, R. C., Little, R. C., Prettyman, T. H., Maurice, S., et al. (2002). Iron abundances on the lunar surface as measured by the Lunar Prospector gamma-ray and neutron spectrometers. *Journal of Geophysical Research*, 107(E12), 5310. https:// doi.org/10.1029/2001JE001530
- LExSWG. (1995). Lunar surface exploration strategy. Lunar Exploration Science Working Group, Final Advisory Group Report. URL: https://www.lpi.usra.edu/lunar/strategies/LExSWG.pdf (Status: 08/16/2018).
- Li, S., Lucey, P. G., Milliken, R. E., Hayne, P. O., Fisher, E., Williams, J.-P., et al. (2018). Direct evidence of surface exposed water ice in the lunar polar regions. Proceedings of the National Academy of Sciences, 115(36), 8907–8912. https://doi.org/10.1073/pnas.1802345115
- McGovern, J. A., Bussey, D. B., Greenhagen, B. T., Paige, D. A., Cahill, J. T., & Spudis, P. D. (2013). Mapping and characterization of nonpolar permanent shadows on the lunar surface. *Icarus*, 223(1), 566–581. https://doi.org/10.1016/j.icarus.2012.10.018
- Metzger, P. T., Anderson, S., & Colaprete, A. (2018). Experiments indicate regolith is looser in the Lunar polar regions than at the lunar landing sites. Paper presented at the 16th Biennial International Conference on Engineering, Science, Construction, and Operations in Challenging Environments, Cleveland, OH.
- Meyerhof, G. G. (1957). The ultimate bearing capacity of foundations on slopes. Paper presented at the Proceedings of the 4th International Conference on Soil Mechanics and Foundation Engineering, London, UK.
- Mitchell, J. K., Carrier, W. D. III, Costes, N. C., Houston, W. N., Scott, R. F., & Hovland, H. J. (1973). Soil mechanics. In Apollo 17: Preliminary Science Report (Vol. 330, pp. 8.1–8.22). Washington, DC: National Aeronatutics and Space Administration.

- Mitchell, J. K., Carrier, W. D. III, Houston, W. N., Scott, R. F., Bromwell, L. G., Durgunoglu, H. T., et al. (1972). Soil mechanics. In Apollo 16: Preliminary Science report (Vol. 315, pp. 8.1–8.29). Washington DC: National Aeronatutics and Space Administration.
- Mitchell, J. K., Houston, W. N., Scott, R. F., Costes, N. C., Carrier III, W. D., & Bromwell, L. G. (1972). Mechanical properties of lunar soil: Density, porosity, cohesion and angle of internal friction. Paper presented at the Proceedings of the Lunar and Planetary Science Conference, Houston, TX.
- Moore, H. J. (1970). Estimates of the mechanical properties of lunar surface using tracks and secondary impact craters produced by blocks and boulders (70-229). Retrieved from http://pubs.er.usgs.gov/publication/ofr70229
- National Research Council (2007). The scientific context for exploration of the Moon. Washington DC: National Academies Press.

Paige, D. A., Siegler, M. A., Zhang, J. A., Hayne, P. O., Foote, E. J., Bennett, K. A., et al. (2010). Diviner lunar radiometer observations of cold traps in the Moon's south polar region. *Science*, 330(6003), 479–482. https://doi.org/10.1126/science.1187726

Paige, D. A., Wood, S. E., & Vasavada, A. R. (1992). The thermal stability of water ice at the poles of Mercury. *Science*, 258(5082), 643–646.
Piquette, M., Horányi, M., & Stern, S. A. (2017). Laboratory experiments to investigate sublimation rates of water ice in nighttime lunar regolith. *Icarus*, 293, 180–184. https://doi.org/10.1016/j.icarus.2017.04.017

Pitcher, C., Kömle, N., Leibniz, O., Morales-Calderon, O., Gao, Y., & Richter, L. (2016). Investigation of the properties of icy lunar polar regolith simulants. Advances in Space Research, 57(5), 1197–1208. https://doi.org/10.1016/j.asr.2015.12.030

Robinson, M. S. & ShadowCam Team 2018. ShadowCam: Seeing in the shadows. Presented at Lunar Polar Volatiles, Laurel, MD. #5028. Sargeant, H. M., Bickel, V. T., Honniball, C. I., Martinez, S. N., Rogaski, A., Bell, S. K., et al. (2019). Beaing capacity calculations for PSR regions of the Moon. Figshare. Dataset. https://doi.org/10.21954/ou.rd.9198167.v2

Schorghofer, N., & Taylor, G. J. (2007). Subsurface migration of H₂O at lunar cold traps. *Journal of Geophysical Research*, *112*, E02010. https://doi.org/10.1029/2006JE002779

Schultz, P. H., Hermalyn, B., Colaprete, A., Ennico, K., Shirley, M., & Marshall, W. S. (2010). The LCROSS cratering experiment. Science, 330(6003), 468–472. https://doi.org/10.1126/science.1187454

- Shean, D. E., Alexandrov, O., Moratto, Z. M., Smith, B. E., Joughin, I. R., Porter, C., & Morin, P. (2016). An automated, open-source pipeline for mass production of digital elevation models (DEMs) from very-high-resolution commercial stereo satellite imagery. *ISPRS Journal of Photogrammetry and Remote Sensing*, 116, 101–117. https://doi.org/10.1016/j.isprsjprs.2016.03.012
- Siegler, M., Paige, D., Williams, J.-P., & Bills, B. (2015). Evolution of lunar polar ice stability. *Icarus*, 255, 78–87. https://doi.org/10.1016/j. icarus.2014.09.037
- Siegler, M. A., Keane, J. T., & Paige, D. A. (2018). Subsurface ice stability on the Moon. Presented at Lunar Polar Volatiles, Laurel, MD. #5038.
- Spudis, P., Bussey, D., Baloga, S., Cahill, J., Glaze, L., Patterson, G., et al. (2013). Evidence for water ice on the Moon: Results for anomalous polar craters from the LRO Mini-RF imaging radar. *Journal of Geophysical Research, Planets*, 66, 1598–1600. https://doi.org/10.1002/jgre.20156
- Terzaghi, K. (1951). Theoretical soil mechanics. London: Chapman And Hall, Limited.
- Warren, T., King, O., Bowles, N., Sefton-Nash, E., Fisackerly, R., & Trautner, R. (2019). The Oxford 3D Thermophysical Model with application to PROSPECT on the Luna-27 lunar lander mission. Presented at the European Lunar Symposium, Manchester, United Kingdom.
- Watson, K., Murray, B., & Brown, H. (1961). On the possible presence of ice on the Moon. Journal of Geophysical Research, 66(5), 1598–1600. https://doi.org/10.1029/JZ066i005p01598
- Williams, J.-P., Paige, D., Greenhagen, B., & Sefton-Nash, E. (2017). The global surface temperatures of the Moon as measured by the Diviner Lunar Radiometer Experiment. *Icarus*, 283, 300-325. https://doi.org/10.1016/j.icarus.2016.08.012
- Zhang, J. A., & Paige, D. A. (2009). Cold-trapped organic compounds at the poles of the Moon and Mercury: Implications for origins. *Geophysical Research Letters*, 36, L16203. https://doi.org/10.1029/2009GL038614
- Zhang, J. A., & Paige, D. A. (2010). Correction to "Cold-trapped organic compounds at the poles of the Moon and Mercury: Implications for origins". Geophysical Research Letters, 37, L03203. https://doi.org/10.1029/2009GL041806